Author's response to reviews

Title: Somatic mitochondrial DNA Alterations In Esophageal Cancer: Significance of Novel Missense and Frameshift Mutations and Alteration in mtDNA Content

Authors:

Duanjun Tan (dt33@georgetown.edu)
Julia Chang (67842@cch.org.tw)
Lingling Liu (liull529@yahoo.com)
Ren-Kui Bai (renkuibai@yahoo.com)
Yu-Fen Wang (82587@cch.org.tw)
Kun-Tu Yhe (10159@cch.org.tw)
Leejun C Wong (ljwong@bcm.tmc.edu)

Version: 6 Date: 21 March 2006

Author's response to reviews:

Re: 1005499293890781, somatic mitochondrial DNA alterations in esophageal cancer

Dear Editorial Team:

We appreciate your acceptance of our manuscript. We have followed the editorial recommendations to make the suggested changes in title, abstract, and other sections of the paper. Regarding the competing interests, the answers to all the questions in financial and non-financial competing interests are "NO".

We have gone through the manuscript formatting checklist and made sure that the formatting of the manuscript is in compliance. We upload the figures separately.

We hope you find this final version satisfactory.

Sincerely yours,

Lee-Jun C. Wong